Author's response to reviews

Title: SUMOylation of sPRDM16 promotes leukemogenesis in acute myeloid leukemia

Authors:

Song Dong (dongsonne@yeah.net)
Jieping Chen (chenjpxn@163.com)

Version: 2
Date: 9 June 2015

Author's response to reviews: see over
Dear Editors,

We would like to submit the enclosed manuscript entitled “SUMOylation of sPRDM16 promotes leukemogenesis in acute myeloid leukemia”, which we wish to be considered for publication in “BMC Cancer”.

No conflict of interest exits in the submission of this manuscript, and manuscript is approved by all authors for publication. This paper has not been published elsewhere in whole or in part. All authors have read and approved the content, and agree to submit it for consideration for publication in your journal. There are no ethical/legal conflicts involved in the article. The research was conducted in accordance with the Declaration of Helsinki and with the Guide for Care and Use of Laboratory Animals as adopted and promulgated by the United National Institutes of Health. All experimental protocols were approved by the Review Committee for the Use of Human or Animal Subjects of Shanghai Jiao Tong University School of Medicine. This manuscript has been edited and proofread by Medjaden Bioscience Limited. RNA-seq data is available for free browsing when requested.

In this work, we evaluated that sPRDM16 SUMOylation contributes to leukemogenesis in acute myeloid leukemia *in vivo* and *in vitro*. I hope this paper is suitable for “BMC Cancer”.

Your consideration for this manuscript is highly appreciated. We look forward to receiving comments from the reviewers. If you have any queries, please don’t hesitate to contact me at the address below.

Thank you and best regards.

Yours sincerely,
Song Dong

Corresponding author:
Name: Jieping Chen
E-mail: chenjpxn@163.com